Author's response to reviews

Title: Systematic Literature Review Assessing Tobacco Smoke Exposure as a Risk Factor for Serious Respiratory Syncytial Virus Disease Among Infants and Young Children

Authors:

Joseph R. DiFranza (Joseph.DiFranza@umassmemorial.org)
Anthony Masaquel (MasaquelA@MedImmune.com)
Amy M. Barrett (abarrett@rti.org)
Ann D. Colosia (acolosia@rti.org)
Parthiv J. Mahadevia (MahadeviaP@MedImmune.com)

Version: 3 Date: 22 May 2012

Author's response to reviews:

We have changed the manuscript to place the Authors' Contributions section after Competing interests. We have also reformatted the Authors' Contributions.